Author's response to reviews

Title: Adoption of New Health Products in Low and Middle Income Settings: How Product Development Partnerships Can Support Country Decision Making

Authors:

William A Wells (william.wells@tballiance.org)
Alan Brooks (alan.brooks@unibas.ch)

Version: 5 Date: 23 March 2011

Author's response to reviews:

One clumsy sentence was reordered to clarify. Several minor typographical issues were fixed.

As we understand this to be a proofing stage, there were no substantive changes made to the meaning of the article text, and there is no need for re-review.